Author's response to reviews

Title: Risk Factors for treatment delay in pulmonary tuberculosis in Recife, Brazil

Authors:

Martinho MAPS dos Santos (martinhoapsantos@hotmail.com)
Maria MFPM Albuquerque (militao@cpqam.fiocruz.br)
Ricardo RAA Ximenes (rximenes@elogica.com.br)
Norma NLCL Lucena-Silva (nlucena@cpqam.fiocruz.br)
Cynthia C Braga (braga@cpqam.fiocruz.br)
Antonio ARL Campelo (rcampelo@nlink.com.br)
Odimariles OMS Dantas (odimariles@globo.com)
Ulisses UR Montarroyos (ulisses@cpqam.fiocruz.br)
Wayner WV Souza (wayner@cpqam.fiocruz.br)
Alexandre AM Kawasaki (kawasaki@hotlink.com.br)
Laura LC Rodrigues (laura.rodrigues@lshtm.ac.uk)

Version: 2 Date: 17 December 2004

Author's response to reviews:

Dear Editor

We apologize for the mistake related to the formatting of the manuscript. According to the instructions a title page comprising the name and information of all authors in the addition to the institutions and postal address of the submitting author was added to the document.

Thanks for your attention

Cynthia Braga